Author's response to reviews

Title: Effects of interval between age at first birth and age at diagnosis on breast cancer survival according to menopausal status: A register-based study in Korea

Authors:

JungSun Lee (gsjslee@gmail.com)
Minkyung Oh (minkyung@busanpaik.ac.kr)

Version: 6 Date: 1 September 2014

Author's response to reviews:

1. We changed the background of this manuscript like that The influence of parity and time interval between age at first pregnancy (AFP) and age at diagnosis on breast cancer survival is not established in the same way as their influence on breast cancer risk. We aimed to investigate the association of time interval or parity with prognosis in pre- and postmenopausal women in Korea.

2. We filled out the STORE checklist. We uploaded the filled-out checklist.